Author's response to reviews

Title: Functional improvements desired by patients before and in the first year after total hip arthroplasty

Authors:

Kristi E Heiberg MrS (k.e.heiberg@medisin.uio.no)
Arne Ekeland Dr (arne.ekeland@mhh.no)
Anne M Mengshoel Professor (a.m.mengshoel@medisin.uio.no)

Version: 4 Date: 21 March 2013

Author's response to reviews: see over
Dear Editor:

We resubmit our revised manuscript for consideration for publication in BMC Musculoskeletal Disorders. The title of the manuscript is:

"Functional improvements desired by patients before and in the first year after total hip arthroplasty". 1021992396800543.

We are grateful for the questions raised by the reviewers and have addressed the concerns in our response letter in the way we considered best.

We look forward to hearing from you.

Yours sincerely,

Kristi Elisabeth Heiberg     Arne Ekeland     Anne Marit Mengshoel
RPT, MSc                           MD, PhD           Professor, PhD, RPT

Corresponding Author
k.e.heiberg@medisin.uio.no